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04 September 2011
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 35 of 2011, which covers the period from 28 August through 03 September 2011.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 113.4 to 176.1 kcfs.  Daily average project tailwater ranged from 10.7’ to 15.5’ msl.  Daily average powerhouse forebay elevation ranged from 73.8’ to 75.6’ msl.  Daily average river temperature ranged from 69 to 71ºF. Secchi disk measurements ranged from 6.0’ to 7.0’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Daily average spill ranged from 1.3 to 94.1 kcfs.
1.3 Unit Operation.  Powerhouse Two is the priority powerhouse.
1.3.1 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.3.2 Unit 17 went out of service at 1010 on 09 August due to generator vibration.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris. 
1.5 Gatewell drawdowns are measured on Tuesdays and more frequently as needed.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS closures occurred this reporting week.
2.1.1 FV 3-7 and 3-9 were closed for cleaning from 0709 until 0800 on 29 August
2.2 STS/VBS inspections.  The following STS/VBS inspection occurred this reporting week. 
2.2.1 Table 1.  STS and VBS inspections.
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	U11 remains out of service.

	12
	8/31
	545
	OK

	13
	8/31
	982
	OK

	14
	8/31
	643
	OK

	15
	8/31
	129
	OK

	16
	8/31
	574
	OK

	17
	8/31
	503
	OK

	18
	8/31
	1015
	OK


2.3 Dewatering and Fish Salvage. No dewatering or fish salvage activities occurred this reporting week.
3 RESEARCH  
3.1 Battelle - survival study.  Monitoring University of Idaho acoustic tagged adult lamprey.   
3.2 CRITFC – AFF.  Annual chinook, coho, sockeye and steelhead sampling in the AFF continues. 
3.3 CRITFC - Sea lion.  Acoustic receivers for sea lion tags remain anchored in the BRZ.  
3.4 NOAA/PSMFC - lamprey.  LPS count network improvements moving forward.  Traps deployed at PH2 monoliths in tailrace and AFF. 
3.5 PSMFC – Smolt Monitoring Program.  Juvenile fish condition and index monitoring continues. Higher temperature sampling protocols were initiated at the beginning of the week when water temperatures exceeded 70ºF. No fallbacks were recorded crossing the separator bars this reporting week. Gas Bubble Trauma (GBT) examinations occur two days per week at the SMF during spill.  Please go to http://www.fpc.org/CurrentDaily/Gbtsum.txt for further details.  
3.6 University of Idaho.  Tagging lamprey in the AFF continues.
3.7 University of Washington – JMF. Juvenile shad collections scheduled to begin in September.

3.8 WDFW - Fish Counting.  Visual counting continues.
3.9 WDFW – Tule Counting. Tule counting began 17 August.
3.10 WDFW – AFF. Sampling for alternative gear study control group began 24 August. 
3.11 Yakama Fisheries – AFF. Coho energy study began 22 August. 
4 FISHWAYS 
4.1 Project biologists inspected 28 August through 03 September.
4.2 Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI)

	9/2
	B-branch entrance
	Diff > 2.0’
	System in local
	Ops adjusted
	

	SPILLWAY

	9/1
	Bay 17
	Open 2.9’
	Gate in local
	Riggers closed
	

	CASCADES ISLAND (CI)

	9/2
	FG 6-18
	Open but indicates closed
	Unknown
	Ops adjusted
	

	WASHINGTON SHORE (WS)

	8/29
	PH2 NUE, NDE diff
	Diff < 1.0’
	System in local
	Ops adjusted
	

	All week
	SDE, NDE
	Will not lower past 3.5’
	Unknown, possible debris in gate slot
	TR
	11-220488

	8/28-30
	DSM2
	SC #2 OOS
	Arm chain broken
	TR
	11-241840


4.2.1 PH1 PCC diffuser FG2-19 is partially open.  Repairs scheduled for 2011/2012 in water work period.  

4.2.2 A new mechanical velocity meter has been fabricated for PH2 and is scheduled for installation.
4.2.3 The SA-24 board that controls the Powerhouse Two adult ladder components is in need of replacement. Work has begun on replacing the PLC to allow for fully automated operation of the PH2 adult fishway.
4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.
Table 3.  Bonneville Fishway Temperatures

	Date
	WA. Shore average
	BI

average
	AFF

average
	SMF

Average*

	8/28
	69.8
	70.1
	69.8
	70.5

	8/29
	69.3
	69.6
	69.4
	70.1

	8/30
	68.8
	69.0
	68.8
	69.6

	8/31
	67.9
	68.1
	68.0
	68.7

	9/1
	67.3
	67.5
	67.4
	68.0

	9/2
	67.3
	67.5
	67.4
	68.1

	9/3
	67.3
	67.4
	67.3
	68.0


*The temperature probe in the SMF may be malfunctioning and is being investigated.

4.3.1 Water temperature in the AFF officially reached 70°F at 1400 hours on 24 August. Temperature restrictions for the AFF were in effect through the end of the reporting week per Appendix G of the FPP. Temperatures can be monitored here: www.nwd-wc.usace.army.mil/tmt/documents/ops/temp/201108.lcol.html
4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 02 September.
4.5.2 10 of 12 floating orifice gates (FOGs) are installed with sea lion barriers in place.  Two gates remain out for maintenance. 
4.5.3 SLEDs were removed from all entrance gates as of 05 July. 
4.5.4 The AFF remains watered up.
4.6 Juvenile Fishways.
4.6.1 DSM2 remains in service. The ERG tripped at 1307 on 30 August. It was closed and the DSM was back in normal operation at 1406. The trip was related to a mechanical brush failure.
4.6.2 The PH1 ice and trash sluiceway (ITS) remains in service.  It was closed from 0710 until 1034 on 23 August, from 0712 until 1651 on 24 August, and from 0719 until 1314 on 25 August to facilitate installation of new booms around FV 3-7, 3-9, and 1-1. Manual chaingates 1A and 1B remain open and are both set to 70’ msl. Automated chaingates 3B, 6C and 10B are open and functioning in auto.
4.6.3 The hydrocannon was returned to service on 06 July.  
4.6.4 The B2CC remains in service.  Five avian lines are broken due to high tailwater and three remain. 
4.6.5 Truck releases of summer out-migrants originating from McNary began on 03 August. Truck releases from Lower Granite, Little Goose and Lower Monumental dams commenced on 17 August.
4.6.6 Spill ended at 0000 hrs on 1 September. Spillbay 17 which must be closed manually was on sill by 1046.  The B2 Corner Collector was closed at 0719 on 1 September.
4.7 Lamprey Fishways.
4.7.1 The Cascades Is. entrance weir remains installed.  The CI LPS is in operation 7 days a week.
4.7.2  The NDE ramp remains in the lowered position.
4.7.3 WA Shore and Bradford Island LPSs remain watered up.
5 CONSTRUCTION
5.1 PH1 Gantry 4.  Work has begun to remove the old crane. Once complete, the new crane will be delivered.
5.2  B2CC Hoist. Construction scheduled to begin fall of 2011.
6 HAZMAT, SPILLS AND CLEANUP. Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 15 June requests summer spill for juvenile fish passage with two treatments of 85 kcfs during the day and 121 kcfs at night or 95 kcfs for 24 hours.  Effective 16 June at 0430 hours and continuing until 31 August at 2400 hours, spill in accordance with the 2011 Fish Passage Plan and the 2011 Summer Fish Operations Plan, Appendix E of the Fish Passage Plan.  
7.2 A teletype sent on 18 August requests that BON pool be operated as follows for treaty fishing: Operate within a 1.5 foot band, hard constraint for periods and times, over the next three weeks. From 0600 hours Monday 22 August - 1800 hours Thursday, 25 August, 2011; from 0600 hours Tuesday 30 August – 1800 hours Friday, 02 September 2011; from 0600 hours Tuesday 06 September – 1800 hours Friday, 09 September 2011.
7.3 A teletype sent on 26 August requests that BON pool be operated as follows for treaty fishing: Operate within a 1.5 foot band, hard constraint for the periods and times, over the next three weeks, as listed: From 0600 hours Monday 22 August – 1800 hours Thursday, 25 August 2011; from 0600 hours Monday 29 August – 1800 hours Friday, 02 September 2011; from 0600 hours Tuesday 06 September – 1800 hours Saturday, 10 September 2011.

7.4 A teletype sent on 26 August requests that on Saturday, 17 September, 2011 from 0800 to 1600 hours, BON forebay be maintained in a 1.5 foot band of 72.0 feet - 73.5 feet as a hard constraint. It is also requested as a soft constraint to maintain the BON pool in a 1 foot band of 72.0 feet -73.0 feet for the same date and times. This operation will assist a clean-up event at the mouth of the White Salmon river to remove abandoned boats and trash that could be washed downstream when Condit Dam is breached in late October.
7.5 A teletype sent on 29 August requests that on Monday, 5 September, outflows from the Bonneville and The Dalles be as low and constant as possible from 0600-1300 to allow for the cross channel swim at Hood River.
JERRY CARROLL
Operations Project Manager
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